Author's response to reviews

Title: Colonic Tuberculosis mimicking Crohn's disease; Report of a case

Authors:

Constantinos Chatzicostas (matrix01gr@yahoo.gr)
Ioannis E Koutouvakis (ktjohn@her.forthnet.gr)
Maria Tzardi (tzardi_maria@yahoo.co.uk)
Maria Roussomoustakaki (rousso@med.uoc.gr)
Panagiotis Prassopoulos (gournick@med.uoc.gr)
Elias A Kouroumalis (kouroum@med.uoc.gr)

Version: 3 Date: 9 May 2002

The suggestion has been accepted and the phrase has been modified (instead of "more than 200" it became several).